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ABSTRACT 

Fluidized bed reactors are increasingly being employed in various solid processing and 

energy industries. With the widening of the scope and complexity of applications, the gamut of 

solids handled has increased manifold. One of the principal fields of its deployment relates to its 

utility as coal gasifiers. Even though industrial coal processing in fluidized bed has been in know-

how for nearly a century now, fundamental knowledge concerning flow hydrodynamics has largely 

impeded its operation and scale up to being mostly empirical. Thus the primary motivation of the 

present thesis is to investigate the underlying hydrodynamics of fluidized beds affecting coal 

gasifiers. 

The first part of the study is focussed on deciphering velocity fields in a binary fluidized bed 

(0.1 mm ID) of standard bed materials. Non-invasive Radioactive Particle Tracking (RPT) was 

employed to measure the solids velocity of a bed with a mixture of glass beads of sizes 0.5 and 2 

mm. The coarser fraction for the present study was varied from 10-40%. Distinct features of solids 

flow were observed with varying bed composition. The mean velocity of finer flotsam (0.5 mm 

particles) fractions were found to achieve higher velocity, which is reduced upon increasing the 

jetsam fraction (2 mm) due to variation in bed voidage.  

The second part of the thesis deals for the first time the determination of velocity fields in a 

fluidized bed of coal-ash mixtures. Suitable tracers were designed and deployed to determine the 

solids flow path of a typical particle through RPT. Experiments were conducted in a cylindrical 

bed of 0.1 m in ID with two inventory compositions, namely, 70% Coal + 30% Bottom Ash 

(Composition I) and 20% Coal + 80% Bottom Ash (Composition II). Comparative analysis of RPT 

measurements yielded overall bed behavior, which was quantified in terms of time-averaged mean 

velocity, solids flux, RMS velocity, and kinetic energy per unit volume of the coal – bottom ash 
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mixture. The mean axial velocity and solids flux profile indicated the influence of strong slugs on 

the solids flow patterns. Further, a three-dimensional transient CFD model was set up to predict 

the time-averaged profiles of coal and ash concentrations. Further, the distributions were compared 

against solids holdup (coal and ash) characteristics measured using γ-ray densitometry, with a 

single collimated source. Similarly, a comparison of measured solids velocity profiles is also 

reported between the CFD-predicted and those measured by RPT. 

The last part of the study dwells on Residence Time Distribution (RTD) measurements and 

its analysis. The experiments were conducted by pulse radiotracer (Au-198) injection in a ‘hot’ 

industrial fluidized bed coal gasifier. The data obtained was treated and analyzed employing a 

network model based on a cascade of perfectly mixed tanks in series. Based on the non-ideality of 

flow, three flow Schemes 1, 2, and 3 were proposed whose RTD function was deduced in the 

Laplace domain. These RTD functions were numerically transformed into the time domain and 

were compared with that measured using radiotracing. It was noted that Scheme 3, which was 

based on the backflow of solids in the bed could satisfactorily predict the actual RTD function. 

Besides, the parameters of the model provided insights into the actual solids flow pattern in the 

gasifier. Furthermore, the model demonstrated its adaptability in analyzing RTD of similar studies 

reported earlier for both single as well as pulse injection of the tracer.  

 

 

  

 



सार 
 द्रवित बिस्तर ररएक्टर विभिन्न ठोस प्रसंस्करण और ऊर्ाा उद्योगों में तेर्ी से कायारत 
हैं। अनुप्रयोगों के दायरे और र्टटलता के विस्तार के साथ, संिाले गए ठोस पदाथों का सरगम 
कई गुना िढ़ गया है। इसकी तैनाती के प्रमुख क्षेत्रों में से एक कोयला गैसीफायर के रूप में इसकी 
उपयोगगता से संिंगित है। िले ही द्रवित बिस्तर में औद्योगगक कोयले का प्रसंस्करण लगिग एक 
सदी से हो रहा है, लेककन प्रिाह हाइड्रोडायनाभमक्स के िारे में मौभलक ज्ञान ने मोटे तौर पर इसके 
संचालन को िागित कर टदया है और ज्यादातर अनुििर्न्य होने का पैमाना है। इस प्रकार ितामान 
थीभसस की प्राथभमक प्रेरणा कोयला गैसीफायर को प्रिावित करने िाले द्रवित िेड के अतंर्नाटहत 
हाइड्रोडायनाभमक्स की र्ांच करना है। 
 अध्ययन का पहला टहस्सा मानक बिस्तर सामग्री के एक िाइनरी द्रिीकृत बिस्तर (0.1 
भममी आईडी) में िेग के क्षेत्र पर डडकॉककंग है। गैर-आक्रामक रेडडयोिमी कण टै्रककंग (RPT) को 
आकार के ग्लास मोर्तयों के भमश्रण 0.5 और 2 भममी के साथ बिस्तर के ठोस िेग को मापने के 
भलए र्नयोजर्त ककया गया था। ितामान अध्ययन के भलए मोटे अशं 10-40% से भिन्न थे। अलग 
बिस्तर संरचना के साथ ठोस प्रिाह की विभशष्ट विशषेताएं देखी गईं। मध्यम िेग (0.5 भममी 
कण) अशंों के मध्य िेग को उच्च िेग प्राप्त करने के भलए पाया गया था, र्ो कक िेड िेपरेर् में 
भिन्नता के कारण र्ेट्सम अशं (2 भममी) को िढ़ाने पर कम होता है। 
 थीभसस का दसूरा टहस्सा पहली िार कोयले की राख भमश्रण के द्रवित बिस्तर में िेग 
क्षेत्रों का र्निाारण करता है। आरपीटी के माध्यम से एक विभशष्ट कण के ठोस प्रिाह पथ को 
र्निााररत करने के भलए उपयुक्त टे्रसर डडर्ाइन और तैनात ककए गए थे। प्रयोगों को दो इन्िेंट्री 
रचनाओं के साथ आईडी में 0.1 मीटर के एक िेलनाकार बिस्तर में आयोजर्त ककया गया था, 
अथाात,् 70% कोयला + 30% िॉटम ऐश (रचना I) और 20% कोयला + 80% िॉटम ऐश 
(रचना II)। RPT मापों के तुलनात्मक विश्लेषण से कुल भमलाकर बिस्तर का व्यिहार हुआ, र्ो 
कक समय-औसतन औसत िेग, ठोस प्रिाह, RMS िेग और कोयले की ऊर्ाा इकाई प्रर्त कोयले 
की मात्रा के िरािर मात्रा में भमलाया गया था। माध्य अक्षीय िेग और ठोस प्रिाह प्रोफाइल न े
ठोस प्रिाह पैटना पर मर्िूत स्लग के प्रिाि को इंगगत ककया। इसके अलािा, एक बत्र-आयामी 
क्षणणक सीएफडी मॉडल की स्थापना कोयला और राख सांद्रता के समय-औसत प्रोफाइल की 
िविष्यिाणी करने के भलए की गई थी। इसके अलािा, वितरण की तुलना एकल ठोस स्रोत के 



साथ सॉभलड होल्डअप (कोयला और राख) विशषेताओं के साथ की गई, जर्सका उपयोग वितरक 
डेंभसटोमेट्री से ककया र्ाता है। इसी प्रकार, सीएफडी-पूिाानुमार्नत और आरपीटी द्िारा मापे गए के 
िीच मापा ठोस िेग प्रोफाइल की तुलना िी ररपोटा की र्ाती है। 
 अध्ययन का अरं्तम िाग रेभसडेंस टाइम डडस्ट्रीब्यूशन (आरटीडी) माप और उसके विश्लेषण 
पर रहता है। प्रयोग गमा ’औद्योगगक द्रियुक्त बिस्तर कोयला गैसीफायर में पल्स रेडडयोटै्रसर (Au-
198) इंर्ेक्शन द्िारा ककए गए थे। प्राप्त आंकडों का इलार् ककया गया और श्रृखंला में पूरी तरह 
से भमगश्रत टैंकों के कैस्केड के आिार पर एक नेटिका  मॉडल को र्नयोजर्त करने का विश्लेषण 
ककया गया। प्रिाह की गैर-आदशाता के आिार पर, तीन प्रिाह योर्नाएं 1, 2, और 3 प्रस्तावित 
की गईं जर्नका आरटीडी फंक्शन लाप्लास डोमेन में घटाया गया था। ये आरटीडी फंक्शन सखं्यात्मक 
रूप से टाइम डोमेन में िदल टदए गए थे और उनकी तुलना रेडडयोटै्रभसगं के उपयोग से की गई 
थी। यह नोट ककया गया था कक योर्ना 3, र्ो बिस्तर में ठोस पदाथों के िैकफ़्लो पर आिाररत 
थी, िास्तविक आरटीडी फंक्शन का संतोषर्नक रूप से अनुमान लगा सकती है। इसके अलािा, 
मॉडल के मापदंडों ने गैसीफायर में िास्तविक ठोस प्रिाह पैटना में अतंर्दाजष्ट प्रदान की है। इसके 
अलािा, मॉडल ने दोनों के भलए पहले से िताए गए समान अध्ययन के आरटीडी के विश्लेषण के 
साथ-साथ टे्रसर के पल्स इंर्ेक्शन का िी प्रदशान ककया। 
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NOTATIONS 

k

dl  Penetration depth of the kth photon in detector crystal 

r Radial coordinate, m 

ni Fractional tracer visit 

umf Minimum fluidization velocity in transition bed,ms-1 

u  Normalized superficial velocity, [-] 

ug Superficial air velocity,ms-1 

z Axial coordinate, m 

Gs Axial mass flux, kg/m2-s 

iv  Instantaneous solid velocity, ms-1 

iv  
Time-averaged solid velocity, ms-1 

'
iv  Fluctuating solid velocity, ms-1 

d Distance traversed in gas-solid medium, m 

e Restitution coefficient,- 

g Gravitational constant, ms-2 

A Source strength (Bq) 

jC
 , 

M

jC  

Calculated and measured count for the detector j using Monte Carlo model for 

detector j  

H, Hw, Hn Expanded Bed height, m 

J Interphase energy exchange 

k Granular Conductivity, kg, m-1s-1 
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P Pressure, Nm-2 

t Time, s 

Greek Letters 

  Fraction of recycle stream in Schemes 1,2 &3, [-] 

  
Momentum exchange coefficient and Fraction of volume of bottom bed, top 

bed, recycle and freeboard in Schemes 1,2 &3, [-] 

2
 

Error in Eq. 3.4 

  
Fraction of backflow in Scheme 3, [-] 

  Phase holdup,- 

  Collision dissipation of kinetic energy and Fraction of bypass stream in 

Schemes 2 &3, [-] 

d  and 

m  

Linear attenuation coefficients of the detector and intervening medium 

between source and detector 

  
Viscosity, kgm-1s-1 


 

Number of photons emitted per disintegration (radioactive source property) 

  
Fraction of total energy spectrum captured by the detector 


 

Error in Eq. 3.3 

  Density,kgm-3 

k
 

Solid angle fraction of the kth photon 

  
Stress, Nm-2 
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  
Granular temperature, m2s-2 

Subscripts 

a Ash 

c Coal 

f air 

i Phase 

s,j Solid phase 

d diameter, µm or mm 

C Concentration, [SI] 

D Dispersion [m2/s] 

E  RTD Function, [1/s] or [1/min] 

Gs Solid Flux, [kg/m2-s] 

M Number of times a tracer/fluid element undergoes recirculation in Schemes 1,2 and 3 

N Total data 

ug Superficial gas velocity [m/s] 

s Complex Laplace domain 

t time, s or min 

B  Bottom Bed 

T  Top Bed 

F  Freeboard 

R  Recycle 

p Particle 

1-3 Schemes 1, 2 and 3 
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Abbreviations 

ADM Axial Dispersion Model 

AR Aspect Ratio 

MC Monte Carlo 

MC-RGA Monte Carlo Real Coded Genetic Algorithm 

RANS Reynolds Averaged Navier Stokes 

RMS Root Mean Square 

RPT Radioactive Particle Tracking 

RTD Residence Time Distribution 
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